Author's response to reviews

Title: Ketoacidosis due to low-carb diet in a non-diabetic lactating female: A case report.

Authors:

Magnus Ekelund (magnus.ekelund@med.lu.se)
Louise von Geijer (louise.vongeijer@skane.se)

Version: 5 Date: 7 September 2015

Author's response to reviews: see over
Dear editor,

Thanks a lot for your valuable comments and suggestions to further improve the manuscript! See below for a point-by-point response to the concerns.

**Editor’s comment:**

“The case report should include the dose of insulin administered and the duration of treatment. Also the amount of fluids given and the rate of administration are needed. Results of kidney function, diabetic autoantibodies, p-cortisol and tests for thyroid function should be given. The patient rate of recovery and follow-up should also be added.”

**Response ( ):**

The requested information has been added in the text.

Magnus Ekelund

MD, PhD